
-----------------------------------------------------------------------------
                Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
        Mon Apr 10 16:45:41 GMT 2006
Year: 06  Doy: 100
Observer: stueben
WEATHER COMMENT: Mon Apr 10 16:48:01 GMT 2006
Dense altostratus overcast, wind southeast 5-10 mph, temp 39F.
___end___
A NEW TAPE HAS BEEN PUT INTO KAIEE DLT DRIVE, Mon Apr 10 19:02:17 GMT 2006
GONG COMMENT: Mon Apr 10 19:02:30 GMT 2006
Performed GONG PM.  Started up ds9 display which replaces the ximtool
display on the terminal in the shelter.
___end___
Mon Apr 10 19:24:55 GMT 2006    CHIP     Start Patrol
WEATHER COMMENT: Mon Apr 10 19:24:27 GMT 2006
Thin to clear hole approaching from south, opened domes.
___end___
Mon Apr 10 19:28:46 GMT 2006    PICS     Start Patrol
PSPT COMMENT: Mon Apr 10 19:35:33 GMT 2006
Observing.
___end___
Mon Apr 10 19:36:15 GMT 2006    PICS     Start Patrol
**PICS PROBLEM**: Mon Apr 10 19:36:19 GMT 2006
PICS started up funny, no Status messages.  Killed Patrol manually 
(the long method with plenty of Cancel this and Cancel that) and
restarted Patrol.
___end___
WEATHER COMMENT: Mon Apr 10 19:38:22 GMT 2006
In hole but hole doesn’t look MK4-friendly at this time.
___end___
**PSPT PROBLEM**: Mon Apr 10 19:56:14 GMT 2006
Crash 10 minutes ago.  Killed and restarted.
___end___
**pics_PROCESSING PROBLEM**: Mon Apr 10 21:57:09 GMT 2006
Appears Mass Store went down.  Both Allen & Eric’s mssrestore processes
are not working.  We managed a couple of days but consequent days
stalled out trying to retrieve data sets from the mass store.
___end___
Mon Apr 10 23:08:04 GMT 2006    CHIP     End Patrol
Mon Apr 10 23:08:20 GMT 2006    PICS     End Patrol
WEATHER COMMENT: Mon Apr 10 23:08:31 GMT 2006
Clouds have become dense, instruments idled.
___end___
NICE CHIP     IMAGE: 2155
NICE PICSLIMB IMAGE: 2152
NICE PICSDISC IMAGE: 2154
Tue Apr 11 00:50:59 GMT 2006    CHIP     Start Patrol



Tue Apr 11 00:51:10 GMT 2006    PICS     Start Patrol
Tue Apr 11 02:22:27 GMT 2006    CHIP     End Patrol
Tue Apr 11 02:22:35 GMT 2006    PICS     End Patrol
COMMENT: Tue Apr 11 02:25:08 GMT 2006
20 man-hrs on site, two observers - Eric and Allen.
___end___
Tue Apr 11 02:26:06 GMT 2006 
        MkIV    
 


